Author’s response to reviews

**Title:** Prevalence rate of undiagnosed Tuberculosis in the community in Ethiopia from 2001-2014: Systematic Review and Meta-analysis

**Authors:**

Balew Arega (balewmlt@gmail.com)
Kelemu Tilahun (ktwu27@gmail.com)
Abrham Minda (abmen7a@gmail.com)
Asnake Agunie (asfitsame@gmail.com)
Getachew Mengistu (gechomlt@gmail.com)

**Version:** 2  **Date:** 15 Jun 2019

**Author’s response to reviews:**

Response to reviewers and editor

Version: 4; Date: 4/06/2019

Herman Van Oye

BMC Public Health Achieve
Subject: AOPH-D-19-00047

Dear editor

Here, we have carefully considered and tried to address the points raised.

Comments and Responses

Reviewer:-

Comments of the Editor-in-Chief

- Comment: 1. Add the period covered by the systematic review in the title
- Response: the suggestion is accepted(line 1-2)
Comment: page 15: typing error Figure for Figure 1. Be consistent and use Figure instead of Fig. (also in the text)

Response: the suggestion is accepted and typing errors corrected(line 381-382)